Author's response to reviews

Title: Inequity in the provision of and access to palliative care for cancer patients. Results from the Italian survey of the dying of cancer (ISDOC).

Authors:

   Monica Beccaro (monica.beccaro@istge.it)
   Massimo Costantini (massimo.costantini@istge.it)
   Domenico Franco Merlo (franco.merlo@istge.it)

Version: 3 Date: 27 April 2007

Author's response to reviews: see over
Dear Editors,

please find enclosed the revised version of our manuscript ‘Inequity in the provision of and access to palliative care for cancer patients. Results from the Italian survey of the dying of cancer (ISDOC)’, accepted for publication in the BMC Public Health.

In this letter, as requested, a point-by-point response to the concerns of the Biomed Central Editorial Production Team. More specifically:

- In the Abstract, now full sentences appear underneath each heading.
- We removed all highlighting from the manuscript text.
- We changed the brackets used for references according to the standard style of the Journal.
- We removed all capitals from headings and sub-headings.
- We removed speechmarks from the sentence on competing interests.
- The Acknowledgements section in now after the Appendix.
- We re-layouted table 2 according to your suggestions.

Yours sincerely,

Dr. Monica Beccaro
Clinical Epidemiology, National Cancer Institute;
Largo R. Benzi, 10, 16132, Genoa (Italy).
Telephone +39 010 5737458; fax +39 (0)10 354103.
Email: monica.beccaro@istge.it